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7/7/2014 Jaroslav Zalesak - AEM 2 

FD: Detector Assembly Progress 

93% cooled down 



FarDet: Data-Taking 

7/7/2014 Jaroslav Zalesak - AEM 3 
Partition 3 – NO longer used, we are installing APDs on last diblock! 

Partition 2                  Jul 7 at 1pm                     Partition 1  

Partition 2 – Diblocks 1,10*, 13, 14 
Diblocks 1-13 cooled APDs 
2nd –ary oil top-up for DiB 10 
Installation of new APDs on last DB 14 

Partition 1 – diblocks 2,3,4,5,6,7,8,9 
and 11, 12. 
Running ~99% active, cooled APDs 
HV at full gain values. 



7/7/2014 Jaroslav Zalesak - AEM 4 

FD: DAQ Uptime 

 Averaging about 85% over 4 weeks and about long-term avaraged 83% in past week. 
 Mon: water leak happened with issue to back up Timing monitor. 
 Tue:  issue with rolling in and back of new version of the Event Display.  
 Wed:  interlock test (for cooling) of  DB-13, afternoon interlock alarm DB-03 (dew point) 

 Sat: water leak followed cooling issues and several DSO scans due to.  



7/7/2014 Jaroslav Zalesak - AEM 5 

FD: POTs delivered/recorded 

 Not good week as one before, efficiency dropped down to long term. 
 We recorded in average 0.87 e18  (from low 1.04) pots/day with good 84% efficiency. 



7/7/2014 Jaroslav Zalesak - AEM 6 

NearDet: Detector Assembly Progress 

APD installed on the whole diblock 3. 
 Some new APDs installed on diblock 2 and MC. 
 We are taking data on 2 partitions: 3rd - (shifters care) FEB-only,  
2nd - diblocks w/APD at full HV, not cooled  (experts care). 



7/7/2014 Jaroslav Zalesak - AEM 7 

FarDet: 
 26.3 of 28 blocks (94%) of FarDet fully-instrumented, 

running with full gain mode. 
 26 of 28  (93%) blocks successfully cooled down. 
 The ‘Physics’ partition is not contiguous due to oil top-up, 

since today we are running with 10 diblocks now. 
 Efficiency of data taking last week was at about long term 

averaged values, we suffered 2 water leaks and cooling and 
interlock issues. 

 
NearDet:  
 APDs installed on DiB-3, continue on 4, total about 50%. 
 We run NearDet over day with full HV on to see clear 3D 

tracks of NuMI events through detector. 

Summary 
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